Author's response to reviews

Title: Identification of a novel CHEK2 variant and assessment of its contribution to the risk of breast cancer in French Canadian women

Authors:

David J Novak (david.novak@mail.mcgill.ca)
Long Qi Chen (drchenlg@hotmail.com)
Parviz Ghadirian (parviz.ghadirian@umontreal.ca)
Nancy Hamel (nancy.hamel@mail.mcgill.ca)
Philip Zhang (phil.zhang@utoronto.ca)
Vanessa Rossiny (vanessa.rossiny@voila.fr)
Guy Cardinal (Guy.Cardinal@hotmail.com)
Andre Robidoux (andre.robidoux.chum@ssss.gouv.qc.ca)
Patricia N Tonin (patricia.tonin@mcgill.ca)
Francois Rousseau (francois.rousseau@mac.com)
Steven A Narod (Steven.Narod@wchospital.ca)
William D Foulkes (william.foulkes@mcgill.ca)

Version: 7 Date: 13 August 2008

Author's response to reviews:

Slight changes to the first paragraph of the introduction.

Two acknowledgements included.

One NB added in proof, at the end of the paper.